Author's response to reviews

Title: Paternal obesity is associated with IGF2 hypomethylation in newborns: results from a Newborn Epigenetics Study (NEST) cohort.

Authors:

Adelheid Soubry (adelheid.soubry@duke.edu)
Joellen M Schildkraut (joellen.schildkraut@dm.duke.edu)
Amy Murtha (amy.murtha@dm.duke.edu)
Frances Wang (frances.wang@dm.duke.edu)
Zhiqing Huang (zhiqing.hung@duke.edu)
Autumn Bernal (abernal3@mac.com)
Joan Kurtzberg (joan.kurtzberg@dm.duke.edu)
Randy L Jirtle (jirtle@radonc.duke.edu)
Susan K Murphy (susan.murphy@dm.duke.edu)
Cathrine Hoyo (catherine.hoyo@dm.duke.edu)

Version: 10 Date: 21 November 2012

Author's response to reviews: see over